CENWP-OP-B                                                                                                                         October 29, 2007
MEMORANDUM FOR CENWP-OP, ATTN: Operations Biologist
SUBJECT: Bonneville Lock and Dam, Fishway and Fish Activities for Week 43 of 2007, which covers the period from 21 through 27 October 2007.
1 OPERATION SUMMARY

1.1 Daily average river flows ranged from 99.5 to 110.2 kcfs.  Daily average project tailwater elevation ranged from 9.4’ to 10.6’ msl.  Daily average powerhouse forebay elevation ranged from 75.1’ to 76.0’ msl.  Daily average river temperature ranged from 57° to 58°F.  Secchi disk measurements ranged from 5.0 to 7.0’.
1.2 Spill Bays 1 and 18, which supply attraction flow for Cascades Island and B-Branch respectively, are included in the spill pattern.  Daily average spill ranged from 0.6 to 1.3 kcfs.   
1.2.1 Spillbay 18 was on seal during the dive operation for Bradford Island cleanup from 0713 until 1707 on 25 October and from 0624 until 1707 on 26 October.
1.3 Unit Operation. Powerhouse 2 is the priority powerhouse.
1.3.1 Unit 1 went out of service at 1551 on 23 October due to governor trouble.

1.3.2 Unit 4 was out of service from 0804 until 1412 on 24 October and from 1407 until 1657 on 26 October for STS installation.

1.3.3 Unit 7 has remained out of service for turbine rehab since 04 June 2007.

1.3.4 Unit 8 has remained out of service for turbine rehab since 12 September 2006.

1.3.5 Unit 11went out of service at 0303 on 15 October for exciter replacement.
1.3.6 Unit 14 has remained out of service since 13 August for re-wedge and FGE modifications.

1.4 Second Powerhouse fish units provide attraction flow for the Washington Shore fish ladder.  The units are placed on standby, as necessary, during nighttime hours to float debris. 
1.5 Gatewell drawdowns are measured on Wednesdays; VBS drawdowns are measured daily on modified units.
2 ACTIVITIES

2.1 AWS Closures.  The following AWS valves were closed this reporting period:  
2.1.1 FV 4-4 was closed from 0700 until 1730 on 25 and 26 October.  
2.1.2 FV 4-3 was closed from 0700 on 25 October until 1730 on 26 October.  (See 4.5.6)
2.2 STS/VBS Inspections.  No STS/VBS inspections occurred this week: 

Table 1.  STS and VBS inspections.  

	Unit
	Date
	Run Hours
	Comments

	11-18
	NA
	NA
	NA


2.3 Dewatering and Fish Salvage.  No dewatering and fish salvage activities occurred this week.
3 RESEARCH  
3.1 CRITFC. Concluded sampling on 19 October.
3.2 NOAA-Gilbreath.  Salt water rearing at the Smolt Monitoring Facility continues.
3.3 PSMFC Smolt Monitoring.  Six salmon and four steelhead were recorded crossing the separator bars this week.  115 juvenile shad collected for researchers.
3.4 Univ. of Idaho.  Received 0 juvenile shad from the SMF this week.

3.5 USGS shad.  Received 115 juvenile shad from the SMF this week.
3.6 WDFW Fish Counting.  Daily visual counting continues at Bradford Island and Washington Shore.  Video monitoring at Cascades Island began on 1 October.
4 FISHWAYS 

4.1 Project biologists inspected daily from 21 through 27 October.  
4.2  Fish Passage Plan Observations.
Table 2.  FPP Out of Criteria items.

	DATE
	LOCATion
	FPP itEMS
	CAUSE
	RESPONSE
	TR#

	BRADFORD ISLAND AND BRANCHES

	10/21, 27
	B-branch diff.
	Reading  <1.0’>
	System in manual
	Operators adjusted
	

	10/21
	B-branch entrance
	S. gate incorrect position for TW
	System in manual
	Operators adjusted
	

	10/22, 27
	B-branch diffusers
	FG 3-27 incorrect position for TW
	System in manual
	Operators adjusted
	

	10/25
	A-branch staff gauge 
	Reading < 1.0’
	
	Operators adjusted
	

	10/27
	B-branch entrance
	N. gate incorrect position for TW
	System in manual
	Operators adjusted
	

	CASCADES ISLAND

	10/21, 27
	CI diff.
	Reading <1.0’>
	System in manual
	Operators adjusted.
	

	10/21
	CI entrance
	N. gate incorrect position for TW
	System in manual
	Operators adjusted.
	

	All week
	CI exit 
	Debris on exit trashrack
	
	Riggers to clean
	08-7320

	All week
	FG6-14
	½ way open
	Electricians need to do some rewiring.
	Winter maintenance item.
	

	WASHINGTON SHORE

	All week
	DSM2 Airburst
	Screens 3, 6 and 9 not consistent.
	Bad controls.
	Will replace during winter maintenance.
	

	All week
	DSM2 PLC
	Not functioning at DSM2
	
	Repairs won’t be made until Dec/Jan
	


4.2.1 PH1 electronic velocity meter is in an eddy.  Due to this placement, the velocity meter often reads less than 1.5 fps.  Readings to meet FPP criteria are taken from the south end mechanical meter.

4.2.2 PH1 PCC diffusers FG2-22A and FG2-19 are closed or partially open.  Repairs scheduled for 07/08.  

4.2.3 FG3-3 stuck partially open.  Repairs are scheduled for winter 07/08.  TR#07-2660

4.2.4 FG3-5 has a damaged shaft and the valve won’t operate.  Repairs are scheduled for winter 07/08.  

4.2.5 FG3-6 stuck partially open.  Repairs are scheduled for winter 07/08.

4.2.6 FG6-10 closed.  The motor was repaired on 15 August.  The shaft and gate have parted ways, however, so the diffuser will not be operational until repaired in winter 07/08.  Only out of criteria when tailwater is above 19.0’ msl.
4.3 Fishway Temperatures.  Measurements are taken three weirs down from the count station at Bradford Island and at the junction of the UMT and the main ladder at Washington shore. Temperatures are in Fahrenheit.
  Table 3.  Bonneville Fishway Temperatures
	Date
	WS average
	BI average

	10/21
	58.4
	57.7

	10/22
	58.5
	58.1

	10/23
	58.8
	58.4

	10/24
	58.8
	58.5

	10/25
	58.4
	58.1

	10/26
	57.9
	57.7

	10/27
	57.0
	56.7


4.4 Zebra Mussel Monitoring.  No signs of colonization were observed this reporting week.  

4.5 Adult Fishways. 
4.5.1 Calibration was checked on 27 October.  
4.5.2 B-Branch and Cascades Island are in manual.
4.5.3 FV3-7 is in manual until the A-Branch diffusers are repaired and operating correctly.
4.5.4 DSM1 continues to run south.  All available orifices are open.  
4.5.5 The UMT remains closed.  Cascades Island exit remains open. 
4.5.6 FV4-3 and FV4-4 was closed from 0700 on 25 October until 1730 on 26 October for dive operations.   On 25 October, the FV4-3 limit switch failed during the closure, it remained closed until crews could reset limits on 26 October.  During hours outside the dive operation, FV4-4 was placed in manual during this time and operated to maintain B-branch entrance criteria.   

4.6 Juvenile Fishways.
4.6.1 DSM2 PLC is OOS.  Estimated time of repair is December or January.

4.6.2 SMF hydrocannons resumed operation on 04 October.  
4.7 Lamprey Fishways.  
4.7.1 The NDE lamprey ramp was raised for winter on 23 October.
4.7.2 Bradford Island and WA shore lamprey passage structures were winterized on 24 October.  
5 CONSTRUCTION
5.1 PH1 ITS Wall Removal.  Plans to install three automated gates.  Optional items include sloping the floor and removing the DSM wall.
5.2 Bradford Island clean-up.  Removal of contaminated sediment currently underway.
5.3 PH2 VBS mods.  Unit 14 is currently OOS.  U18 will follow.  U17 VBS monitoring equipment will be installed at this time as well.

5.4 TIE crane.  The TIE crane boom repair is on hold until funding is secured.

6 HAZMAT, SPILLS AND CLEANUP.   Nothing to report.
7 TELETYPES AND OTHER NOTEWORTHY ITEMS
7.1 A teletype sent on 07 June requests that all Lower Snake and Columbia River FCRPS projects operate all turbine units within the 1% best efficiency standard during the best efficiency operating period (April 1 through October 31).  Operation outside of 1% is allowed for maintenance, start up/shut down and other instances as outlined in the Fish Passage Plan.  Operation outside of 1% is NOT allowed for any other reason unless a power emergency has been declared.
JAMES R. MAHAR, P.E.

Operations Manager
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